Author's response to reviews

Title: Continuing medical education in Turkey: efforts of the Turkish Medical Association

Authors:

hakan yaman (hyaman@sdu.edu.tr)

Version: 2 Date: 12 Jun 2002

Dear Editor,

following changes has been made to the text:
-the abstract was already 172 words long, therefore the text has not been further shortened.
-the abstract has been structured
-abbreviations in the abstract have been removed
-the main text has been structured according to the headings background, discussion and summary
-competing interests has been declared, stating none declared.
-table in the attachment has been removed. the table is now in the text.
-instructions for authors have been checked again.